Author's response to reviews

Title: The influence of ethnic group composition on focus group discussions

Authors:

Nan Greenwood (Nan.Greenwood@sgul.kingston.ac.uk)
Theresa Ellmers (t.ellmers@gmail.com)
Jess Holley (jessholley2108@gmail.com)

Version: 5 Date: 6 September 2014

Author's response to reviews:

Changes are highlighted in yellow.
As stated in the earlier document, on page 11 line 9, all participants were, of course, given pseudonyms to ensure they were not identifiable.
We have changed the Findings section to Results
We have added a clearer consent section and stated that a copy of the consent form is available if needed.
I hope these changes address your concerns.
Nan Greenwood
We have added a sentence to say that the reporting adheres to the RATS guidelines.
We believe the manuscript follows the BMC format.